Author's response to reviews

Title: Determinants of ANC adequacy in rural versus urban: results from two health and demographic surveillance sites in Vietnam

Authors:

Toan K Tran (tktoan@yahoo.com)
Karin Gottvall (karin.gottvall@ki.se)
Hinh D Nguyen (duc_hinh@yahoo.com)
Henry Ascher (henry.ascher@nhv.se)
Max Petzold (max.petzold@nhv.se)

Version: 2 Date: 1 July 2011

Author's response to reviews: see over
To BMC Health Services Research Journal

Dear Editor,

Enclosed is my article “Determinants of ANC adequacy in rural versus urban: results from two health and demographic surveillance sites in Vietnam” for publishing consideration in your Journal.

The article addresses the inequality in access to and use of antenatal care between different socioeconomic groups in urban and rural settings. This is an important issue under the context of fast economic development in Vietnam but the knowledge is still limited. The study was conducted at two Health and Demographic Surveillance Sites, where the women were followed over time during the pregnancy.

This is original article and it has not been published by or submitted to any other journal. There are no conflicts of interest for any of the authors.

Thank you for your consideration.

Yours sincerely

Tran Khanh Toan